Author's response to reviews

Title: Population-based laboratory surveillance for Giardia sp. and Cryptosporidium sp. infections in a large Canadian health region.

Authors:

Kevin B Laupland (Kevin.laupland@cls.ab.ca)
Deirdre L Church (deirdre.church@cls.ab.ca)

Version: 6 Date: 8 September 2005

Author's response to reviews:

The following formatting changes that were requested have been completed:
1. One corresponding author listed on the title page
2. Affiliations - Each affiliation is written in full and the format followed
3. Figures - The four figures were cropped to minimise white space.